Author's response to reviews

Title: Unexpected combination of acute croup and myocarditis: Case report

Authors:
George Briassoulis (ggbriass@otenet.gr)
Athina Tsorva (briassg@hotmail.com)
Emmanuel Agapitos (eagapit@med.uoa.gr)
John Papadatos (briassg@hotmail.com)

Version: 4 Date: 24 May 2005

Author’s response to reviews:

Country - Please remember to include the country at the end of the affiliation details.

It was included Hellas is the official name of Greece. Now I changed it to Greece.

Abstract- Please change the first sentence under 'case presentation' to read 'A three and a half year old girl...'. Please remember to also revise the abstract details on the submission system accordingly. Please also make this change on page 4 of the manuscript.

Changed

Highlighting- Please remove any red highlighting form the text.

No red highlighting was found

Additional files - Please delete the additional files that contains your response to our decision letter and waiver requests

Deleted